Author’s response to reviews

Title: A family based tailored counselling to increase non-exercise physical activity in adults with a sedentary job and physical activity in their young children: Design and methods of a year-long randomized controlled trial

Authors:
Taija Finni (taija.finni@jyu.fi)
Arja Sääkslahti (arja.saakslahti@jyu.fi)
Arto Laukkanen (arto.laukkanen@jyu.fi)
Arto Pesola (arto.j.pesola@jyu.fi)
Sarianna Sipilä (sarianna.sipila@jyu.fi)

Version: 3 Date: 19 December 2011

Author’s response to reviews:

Dear Editor,

We noted few typos and made the following changes:

E-mail of Sarianna Sipilä is sarianna.sipila@jyu.fi
kindergarten -> kindergartens
in -> into
used abbreviation FMS
added the word Baseline
Bolded the title of www pages in the list of references.

Sincerely,
Taija Finni